Author’s response to reviews

Title: Effects of carvedilol treatment on cardiac cAMP response element binding protein expression and phosphorylation in acute coxsackievirus B3-induced myocarditis

Authors:

Ge Li-Sha (gelisa@sina.com)
Chen Yi-He (heifeng1980@sina.com)
Zhou Na-Dan (zhounadan@163.com)
Zhang Teng (zt1985@sohu.com)
Li Yue-Chun (liyuechun1980@sina.com)

Version: 3  Date: 10 November 2013

Author’s response to reviews:

Dear Editor:

Thank you for your kind letter of November 09, 2013. We have revised the entire manuscript entitled “Effects of carvedilol treatment on cardiac cAMP response element binding protein expression and phosphorylation in acute coxsackievirus B3-induced myocarditis” (MS: 1419026411105962) from start to finish in accordance with the comments of the reviewers. We acknowledge your help and the reviewer’s comments and suggestions very much, which are very valuable in improving the quality of our manuscript.

Here below is our description on revision according to the reviewers’ comments.

Reviewer 1 (Andreas Henke)
1. The reviewer’s comment: Basically I agree with the answers to the comments of the reviewer.
   However, please give a much more precise description of the virus titrations of heart tissue (page 8 lines 19-21).
   The authors’ Answer: Corrected accordingly.

2. The reviewer’s comment: Please have a look at your Figure 10B. Something is wrong with the labeling.
   The authors’ Answer: Corrected accordingly.

Reviewer 2 (Rudin Pistulli)
   The reviewer’s comment: The authors have answered accordingly to my previous comments and have made the required amendments. I have no further comments and see the article now fit for publication.

Thank you and all the reviewers for the kind advice.
Sincerely yours,
Li yue-chun
2013.11.10